CIS 500 Software Foundations Homework Assignment 4

Lambda Calculus Due: Monday, September 29, 2003, by noon Submission instructions: Same as last time.

1 Exercise 5.2.4 in TAPL. 2 Exercise 5.2.5 in TAPL. 3 Exercise 5.2.7 in TAPL. 4 Exercise 5.2.8 in TAPL. 5 Exercise 5.2.11 in TAPL.

6 Debriefing

1. Are you in a study group? 2. If so, who else is in it?

3. How many hours did you spend on this assignment? 4. Would you rate it as easy, moderate, or difficult?

5. Did you work on it mostly alone, or mostly with your study group? 6. How deeply do you feel you understand the material it covers (0%–100%)?

If you have any other comments, we would like to hear them; please send them directly to bcpierce@cis.upenn.edu to make sure they don’t get lost in a pile of homeworks!

All of the solutions can be found in the back of the book.